Author's response to reviews

Title: Elevated 14,15- epoxyeicosatrienoic acid by increasing of cytochrome P450 2C8, 2C9 and 2J2 and decreasing of soluble epoxide hydrolase associated with aggressiveness of human breast cancer

Authors:

Xiaolong Wei (weixiaolonghh@126.com)
Donghong Zhang (donghong1228@163.com)
Xiaowei Dou (daviddow@126.com)
Na Niu (naniubjmu@gmail.com)
Jingwen Bai (baijingwen666@126.com)
Wenhe Huang (huangwenhe2009@163.com)
Guojun Zhang (guojiang@yahoo.com)

Version: 6 Date: 30 October 2014

Author's response to reviews: see over
Dear Dr. Solera,

Thank you very much for providing us with the opportunity to revise our manuscript (MS: 1590356043128170) entitled “**Elevated 14,15-EET by increasing of cytochrome P450 2C8, 2C9 and 2J2 and decreasing of soluble epoxide hydrolase associated with aggressiveness of human breast cancer**”. We have carefully read the editorial comments and have included a conclusion session accordingly. The revisions are highlighted in the manuscript and I hope the revised version will be acceptable for publication in *BMC Cancer*.

I am looking forward to hearing from you soon.

Sincerely,

Guo-Jun Zhang, MD, PhD
The Breast Center, Cancer Hospital of Shantou University Medical College. 7 Raoping Road, Shantou, Guangdong Province, 515031, PR China.
E-mail: guoj_zhang@yahoo.com;
Tel: +86-754-88556826, Fax: +86-754-88556826.